Author's response to reviews

Title: Sleep duration of underserved minority children in a cross-sectional study

Authors:

William W Wong (wwong@bcm.edu)
Christina L Ortiz (tina.ortiz@houstontx.gov)
Debra Lathan (debra.lathan@houstontx.gov)
Louis A Moore (louis.moore@houstontx.gov)
Karen L Konzelmann (dkkontex@gmail.com)
Anne L Adolph (aadolph@bcm.edu)
E. O'Brian Smith (esmith@bcm.edu)
Nancy F Butte (nbutte@bcm.edu)

Version: 3 Date: 26 June 2013

Author's response to reviews: see over
June 26, 2013

The BMC Public Health Editorial Team

RE: MS 1651376446901367 second revision

**Sleep duration of underserved minority children in a cross-sectional study**

William W Wong, Christina L Ortiz, Debra Lathan, Louis A Moore, Karen L Konzelmann, Anne L Adolph, E. O'Brian Smith and Nancy F Butte

Dear Dr. Draper:

Our responses to the reviewers are as follows:

**Reviewer 1:**
Reviewer 1 has no further comment on the revised manuscript. Therefore, no response is necessary.

**Reviewer 2:**
The sentences comparing the Actical measurements with the wrist-worn accelerometers in the Methods section have been removed from the revised manuscript so that the messages will not be repeated.

We hope our responses have adequately addressed the comments of the reviewers and look forward to a favorable response from the Editor.

Sincerely yours,

William W. Wong, Ph.D.